Answer to reviewer 05468066

**Conclusion:** Minor revision

**Scientific Quality:** Grade B (Very good)

**Language Quality:** Grade B (Minor language polishing)

The authors have written this manuscript on an interesting topic. There are no major issues I can point to. However, further English editing could improve readability of this manuscript.

Dear Editor,

We do thank deeply our reviewer for his kind comments regarding our Ms. As requested, it has been read again by an English native friend of ours who made slight modifications. We hope that our Ms is now ready to be published in WJGO.

Best regards

Sylvain Loric